Computer Graphics Brian Curless, 318 Sieg CSE 457, Autumn 1998 Head TA: Craig Kaplan

Homework 0: Questionnaire

Received: Monday, 28 September 1998 Due: Monday, 5 October 1998

Name:

Year:

Major:

Photo or Self-portrait

1. What got you interested in computer graphics?

(over)

2. Tell me about any prior experience you might have with graphics. . . .

3. What are you expecting from this class?

4. Please email a pointer to your home page to Craig Kaplan (csk@cs) so he can link it into the class’s home page.